Author's response to reviews

Title: Weight and weight change following breast cancer: evidence from a prospective, population-based, breast cancer cohort study

Authors:

Dimitrios Vagenas (dimitrios.vagenas@qut.edu.au)
Tracey DiSipio (t.disipio@qut.edu.au)
Diana Battistutta (d.battistutta@qut.edu.au)
Wendy Demark-Wahnefried (demark@uab.edu)
Sheree Rye (sa.rye@qut.edu.au)
John Bashford (jbashford@hoca.com.au)
Chris Pyke (c_pyke@mc.mater.org.au)
Christobel Saunders (christobel.saunders@uwa.edu.au)
Sandra C Hayes (sc.hayes@qut.edu.au)

Version: 5 Date: 8 January 2015

Author's response to reviews:

Dear Dr Downing

Ref: MS: 3653313851287952

Weight and weight change following breast cancer: evidence from a prospective, population-based, breast cancer cohort study

Thank you for your recent correspondence regarding the abovementioned manuscript for publication in BMC Cancer. In accordance with the Editor's Request, the authors have addressed the following comments in the revised manuscript:

• A statement on ethics now appears in the Methods section of the manuscript (see page 4, lines 12-15).

We trust that these revisions meet with your approval and look forward to your reply.

Yours sincerely

Dr Tracey DiSipio